

Limnology and Oceanography Letters 2024 y Letters published by Wiley Periodicals LLC

© 2024 The Author(s). Limnology and Oceanography Letters published by Wiley Periodicals LLC on behalf of Association for the Sciences of Limnology and Oceanography. doi: 10.1002/lol2.10446

SYNTHESIS

Future Arctic: how will increasing coastal erosion shape nearshore planktonic food webs?

Gabriel A. Juma ,^{1,2}* Cédric L. Meunier ,¹ Emily M. Herstoff, Anna M. Irrgang, Michael Fritz, Cedric L. Meunier, Emily M. Herstoff, Anna M. Irrgang, Michael Fritz, Cedric L. Meunier, Hugues Lantuit, Index and Level and Cedric L. Meunier, Index and Anna M. Irrgang, Anna M. Irrgang, Anna M. Irrgang, Cedric L. Meunier, Cedric L. Meunier, Index and Cedric L. Meunier, Anna M. Irrgang, Anna M. Irrgang, Anna M. Irrgang, Cedric L. Meunier, Cedric L.

¹Alfred-Wegener-Institut, Helmholtz-Zentrum für Polar- und Meeresforschung, Biologische Anstalt Helgoland, Helgoland, Germany; ²FB2, University of Bremen, Bremen, Germany; ³Biology Department, St. Francis College, Brooklyn, New York, USA; ⁴Alfred-Wegener-Institut, Helmholtz-Zentrum für Polar- und Meeresforschung, Potsdam, Germany

Scientific Significance Statement

Arctic permafrost coasts constitute 34% of the Earth's coasts, and they are warming nearly four times faster than the global average. This rapid warming erodes Arctic coasts which may have important implications for the functioning of coastal ecosystems, moreso the productivity of planktonic organisms that form the basis of the aquatic food web. Here, we explore how Arctic coastal erosion may impact nearshore primary production and we evaluate implications for the functioning of planktonic food webs. We provide evidence that higher turbidity associated with increased erosion of permafrost coasts is likely to shift the balance between phytoplankton and bacterial production, which may lower zooplankton biomass, with negative consequences for higher trophic levels. Our findings lay out the path for future work to investigate the permafrost-carbon feedback, and how increasing Arctic coastal erosion will affect the microbial food webs.

Abstract

Arctic regimes. Currently, warming accelerates the erosion of permafrost coasts and the associated discharge of sediment, carbon, and nutrients into the Arctic Ocean. However, the impacts of coastal erosion on planktonic food webs remain understudied. We aimed to (1) understand how coastal erosion impacts nearshore carbon, nutrient, and light regimes; (2) investigate the effects on primary production and energy transfer; and (3) predict how increased erosion will impact the productivity of consumers, and the overall food web interactions. We found that sediment discharge increases turbidity (darkening). This darkening is expected to hamper phytoplankton productivity, while additional carbon input will provide bacteria with direct energy sources, and shift the balance between basal autotrophic and heterotrophic production. Since the heterotrophic pathway has a lower efficiency, its dominance might negatively affect mesozooplankton. Increased Arctic coastal erosion might therefore influence planktonic food webs by changing mechanisms of energy mobilization and transfer to higher trophic levels.

*Correspondence: gabriel.juma@awi.de

Associate editor: Dick van Oevelen

Author Contribution Statement: All authors contributed extensively to the work presented in this paper. GAJ, CLM, and MB designed the manuscript; GAJ wrote the manuscript which was revised and improved by CLM, EMH, AMI, MF, CW, HL, IVK, and MB.

Data Availability Statement: Data sharing is not applicable to this article as no new data were generated or analyzed in this study.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Juma et al.

In the last four decades, the Arctic has warmed nearly four times faster than the global average (Lim et al. 2020; Rantanen et al. 2022). This phenomenon has increased the susceptibility of Arctic permafrost coasts to thaw and erode, leading to an estimated increase in erosion rates from around 0.18 m yr^{-1} up through the mid-1970s (Nielsen et al. 2022), to 0.5 m yr^{-1} by 2012 (Lantuit et al. 2012). Accelerating coastal retreat is caused by a combination of many different factors, which vary locally and regionally. The main environmental drivers leading to faster erosion are longer open water periods, more severe, frequent, and longer-lasting storms, rising sea levels, and higher water and air temperatures (Irrgang et al. 2022). These environmental drivers all result in greater wave impacts and less resilient shorelines (Casas-Prat and Wang 2020; Irrgang et al. 2022). Since Arctic permafrost coasts constitute 34% of Earth's coastlines, and since onethird of settlements built on permafrost are close to the coast (Lantuit et al. 2012; Fritz et al. 2017; Ramage et al. 2021), increased erosion in the Arctic has broad consequences for the functioning of coastal ecosystems, as well as for the sustainability of livelihoods and infrastructure (Fritz et al. 2017).

Permafrost coastal erosion mobilizes sediments, nutrients, and carbon, which are subsequently transported from the land to the ocean, and are predicted to greatly alter Arctic planktonic food webs by changing light and nutrient regimes. Currently, coastal erosion is estimated to release an average of 15.4 Tg (10^{12} g) of total carbon, and 1.6 Tg yr⁻¹ of nitrogen into the Arctic Ocean (Terhaar et al. 2021). However, the specific effects of carbon and nutrient input on coastal planktonic food webs are murky, due to high variabilities, interdependencies, and interactions among watersheds and in the coastal zone. Tank et al. (2012) posed that nitrogen from land only supports a small proportion of photosynthesis in the Arctic Ocean, while Terhaar et al. (2021) estimated that coastal erosion sustains about one-fifth of net primary production. Further, there is paucity of information on the impacts of terrigenous material on turbidity, and the balance between phytoplankton and bacterioplankton energy pathways in the nearshore zones (zones shallower than 20 m water depth). These zones account for 7.5% of the Arctic Ocean (Fritz et al. 2017), a much greater proportion compared to other oceans. They also play a major role in Arctic biogeochemical cycling and primary production by determining the fate of the material released from terrestrial ecosystems (Fritz et al. 2017). The increased introduction of terrestrial material into the nearshore zone might lead to the darkening of coastal waters (increased turbidity due to high concentrations of suspended material) (Wegner et al. 2015), which will negatively affect primary production in those zones (Lindemann et al. 2016; Szeligowska et al. 2021). Additionally, intensifying Arctic coastal erosion may affect planktonic community structure and the overall food web functioning in the nearshore zone (Calbet 2008; Vrede et al. 2011; Schmoker et al. 2013).

The foundation of the Arctic pelagic food web is primary production by single-celled phytoplankton and sea ice algae. These autotrophs use light and nutrients to store energy in the form of compounds like polysaccharides, which can later be passed to higher trophic levels via consumption (Bluhm and Rolf 2008; Murphy et al. 2020). In Arctic coastal regions, both underwater light regimes and nutrient supplies are affected by sea ice dynamics, the circulation of water masses, and seasonal permafrost thawing (Popova et al. 2010; Ardyna et al. 2017). Arctic coastal zones are typically characterized by strong vertical and horizontal stratification in the water column and low nutrient levels in the surface layers. However, episodic nutrient inputs can trigger phytoplankton blooms that may account for up to half of the regional production within a season (Mundy et al. 2009). This autotrophic primary production can also be supported by bacterioplankton through heterotrophic production. Indeed, bacterioplankton consume dissolved and particulate organic matter and hence play a key role in remineralizing processes (Piontek et al. 2021). Consumption of autotrophic and heterotrophic production by microzooplankton (< 200 μ m in equivalent spherical diameter) and mesozooplankton (between 0.2 and 20 mm) (Negrete-García et al. 2022), subsequently passes energy and nutrients to higher-order consumers, such as planktivorous fish. The top of Arctic pelagic food webs is characterized by fish species like cod and Arctic char, as well as mammals such as seals and whales, upon which local subsistence fishers and hunters rely (Zeller et al. 2011; Galappaththi et al. 2019). Given that the structure and functioning of Arctic pelagic food webs depend on the interplay between nutrients and light availability (Polimene et al. 2022), changing environmental conditions have the potential to alter the ecosystem services provided to local communities.

While the link between Arctic coastal erosion and primary productivity has been studied to some extent (Sipler et al. 2017; Terhaar et al. 2019, 2021; Polimene et al. 2022), the implications for the balance between autotrophic and heterotrophic energy pathways, and the overall food web efficiency in the nearshore zone, remain largely untested. With ongoing warming, coastal erosion fluxes in the Arctic have the potential to increase by an order of magnitude by 2100 (Rantanen et al. 2022). Taken together, the increase in organic carbon (OC) and the decrease in light availability will have consequences for planktonic energy pathways, but the exact ways these factors interact remain unclear. Understanding the cascading effects of increasing Arctic coastal erosion on the nearshore food web is crucial for conservation efforts and the implementation of climate change mitigation measures, as human communities in the Arctic depend on local marine biological resources to a large extent (Fritz et al. 2017; Galappaththi et al. 2019).

In this review, we assess the potential effects of Arctic coastal erosion on planktonic community structure and food web interactions in nearshore environments. We focus on (1)

understanding how coastal erosion impacts carbon, nutrient, and light availability in the nearshore zones; (2) investigating how these changes could affect autotrophic and heterotrophic production, and energy transfer to higher trophic levels; and (3) understanding how the increase in Arctic coastal erosion will impact the productivity of consumers, and the overall food web interactions and efficiency.

Arctic coastal erosion and its impacts on carbon, nutrients, and light distribution

Arctic coastal systems have a high geomorphic variability (Lantuit et al. 2012), and include lithified and unlithified coasts. Lithified coasts consist of solid rocks and show little erosion. They are characterized by a mix of fjords, pocket beaches, bluffs, and low-lying rocky shorelines (Overduin et al. 2014). Unlithified coasts are composed of unconsolidated sediments and are cemented and solidified by ground ice. They are characterized by ice-rich permafrost bluffs (Lantuit et al. 2012) and are mostly found in Canadian, U.S., and Russian coastal margins, making up two-thirds of the coasts facing the Arctic Ocean (Lantuit et al. 2012; Farquharson et al. 2018; Kupilik et al. 2020; Lim et al. 2020). Permafrost areas comprise a seasonally active layer that thaws during the warm season. Underneath the active layer lies perennially frozen ground that contains substantial amounts of ice. Erosion of the active layer makes Arctic coasts exceptionally vulnerable to degradation upon thaw and mechanical abrasion when in contact with seawater.

Unlithified permafrost coasts store large amounts of OC that are increasingly eroded into the Arctic Ocean. OC stored in permafrost and the active layer above it across the Northern Hemisphere is about 1,300,000 Tg, with an uncertainty range of \sim 1,100,000 Tg to \sim 1,500,000 Tg, which exceeds the amount of carbon stored in all other soil regions of the world (Hugelius et al. 2014). Warming accelerates permafrost thawing and erosion (Cohen et al. 2014; Fritz et al. 2017; Underwood et al. 2019). Compared to stabilized permafrost coasts (Fig. 1a), coastal areas with retrogressive thawing slumps (Fig. 1b), actively eroding cliffs (Fig. 1c), active layer detachment, and low bluffs, are substantially degraded (Grotheer et al. 2020; Tanski et al. 2021). Under the RCP 8.5 scenario, these phenomena are projected to decrease near-surface permafrost areas by $69 \pm 20\%$ by 2100 (Slater and Lawrence 2013). The erosion of the Arctic coast is further enhanced by sea level rise and the intensification of storms (Atkinson 2005). Even though seasonal coastal erosion is a natural process, current global warming and the associated increase in extreme weather events intensify sediment and carbon transport from the land (Wegner et al. 2015). While geomorphological studies on coastal erosion processes are manifold, little is known about how these processes influence carbon bioavailability in nearshore areas, and the subsequent uptake of OC by planktonic communities in the Arctic Ocean.

Filling this knowledge gap on OC transformation and uptake is fundamental in understanding the biogeochemical pathways of OC in the warming Arctic.

Coastal erosion is a natural process resulting from wind and wave action and has always caused substantial OC and nutrient fluxes into the coastal Arctic Ocean. For example, the pan-Arctic average loss from coastal erosion has been estimated at 6.9 Tg C yr⁻¹ during the period 1850–1950 (Mcguire et al. 2009; Nielsen et al. 2022). However, these rates have increased up to 15.4 Tg C yr⁻¹ in 2021 (Wegner et al. 2015; Terhaar et al. 2021). The increase in terrestrial input could drastically impact fluxes of key elements like carbon, nitrogen, and phosphorus into and within the coastal ecosystems and their feedback on nearshore local communities and food webs. This raises the question of the exact consequences in the nearshore zones.

Coastal erosion will increase organic matter supply. The Arctic nearshore system receives considerable amounts of particulate organic carbon (POC) and, to a smaller degree, dissolved organic carbon (DOC). For example, on the Yukon coast, POC : DOC is supplied in the ratio of 900 : 1 (Tanski et al. 2016). These organic compounds are derived from remnants of animals and plants that have accumulated in the permafrost over millennia (Barker et al. 2006). Increased contact between organic soil horizons and water through thawing processes enhances organic matter mobilization into the water column or leaching from the sediment and soils, leading to higher OC concentrations in nearshore systems (Vonk et al. 2013; Jong et al. 2020). Additionally, the small amount of OC stored in ice wedges and other ground ice also gets released during the thawing of ice-rich permafrost (Fritz et al. 2015; Tanski et al. 2016).

Permafrost erosion also has the potential to increase nutrient transport to coastal waters (Thompson et al. 2012). For instance, an estimate of 1.6 Tg N yr⁻¹ of total nitrogen almost entirely in particulate organic form is delivered through coastal erosion (Terhaar et al. 2021). This input is particularly relevant since Arctic systems are nitrogen-limited. The deficiency of dissolved inorganic nitrogen (DIN) in the Arctic waters occurs due to the connectivity of the Arctic Ocean to other oceans, and its circulation (Tremblay et al. 2015). DIN is lost within the Arctic Ocean through denitrification on Arctic shelves while additional surface DIN inventories are depleted after ice retreat, leading to nutrient limitation of primary production during the rest of the season when light is sufficient. In addition to nitrogen, phosphorus (P) is also supplied by permafrost thaw, river runoff, and ice melting in the Arctic environment (Hawkings et al. 2016). Climate-related changes to permafrost erosion, therefore, have particular significance in increasing nutrient supply into the Arctic Ocean.

Unlike carbon and nutrients, light availability might be reduced in the Arctic nearshore systems (Fig. 1d). Allochthonous material originating from permafrost erosion brings an array of humic compounds that significantly



Fig. 1. Examples of different coastal landforms along the Yukon coast in the western Canadian Arctic. Stabilized permafrost coasts show little or no erosion (**a**). However, slump-affected sites (**b**) and coasts with actively eroding cliffs (**c**) exhibit higher erosion and substantial transfer of sediment and OC into the water column. The discharge of the eroded material increases turbidity in the nearshore zones (**d**). Photos were taken by C. Weber in July 2022.

attenuate light and reduce the depth of the euphotic zone (Lefébure et al. 2013; Brett et al. 2017; Mann et al. 2021; Szeligowska et al. 2021). For example, between 2003 and 2020, the attenuation of light in the coastal Arctic Ocean with depth < 100 m, increased by 22% (Singh et al. 2022). Similarly, high sediment loads from Arctic rivers lead to a reduction in light penetration throughout the water column (Wegner et al. 2015), a major consequence given that the Arctic Ocean receives more than 10% of the global river discharge (McClelland et al. 2012). Changes in long-term river discharge into the Arctic Ocean have also been observed, with a 7% increase recorded between 1936 and 1999 (Peterson et al. 2002). The river-driven brownification occurs mainly in spring and early summer. This is linked to maximal riverine runoff, and the highest absolute sediment flux into the ocean (Golubeva et al. 2019). In late summer, large loads from

coastal erosion drive turbidity (Terhaar et al. 2021), whereas in winter, little or no runoff occurs due to the freezing of the land and the rivers. Discharge of freshwater either directly from land or through rivers also lowers surface salinity (Park et al. 2023), thereby increasing surface layer stratification. Because climate change is elongating and intensifying, permafrost thaw, coastal erosion, and seasonal river runoff are expected to further reduce light intensity (Brett et al. 2017; Mann et al. 2021; Szeligowska et al. 2021).

Consequences of increasing coastal erosion for the nearshore planktonic food web

Elevated inputs of permafrost material along the Arctic coast are expected to (1) intensify by 70% to 150% by 2100 in zones already receiving high terrestrial input (e.g., estuaries

and nearshore areas) (Nielsen et al. 2022), (2) increase in zones that until now received little terrestrial input, and (3) increase the overall coastal area influenced by terrestrial inputs. These processes may have wide-ranging consequences for Arctic marine ecological dynamics, including influencing species interactions, productivity, and energy transfer. Here, we explore the potential impacts of increasing coastal erosion on plankton productivity and food web interactions, focusing on the poorly studied yet highly relevant lower trophic levels.

Phytoplankton

Two main processes are associated with the influence of erosion on nearshore primary production: a decrease in light availability, and an increase in carbon and nutrient concentrations. Although nitrogen is limited in the Arctic Ocean, coastal erosion mobilizes nutrients and supports 28-51% of net primary production in the coastal zone (Terhaar et al. 2021). However, terrestrial input influences nearshore marine habitats by decreasing the extent of the euphotic zone (McGovern et al. 2020; Szeligowska et al. 2021). Given the increase of up to 0.01 $K_d^{\text{PAR}[m-1 \text{ yr}-1]}$ in light attenuation between 2003 and 2020 (Singh et al. 2022), turbidity can be expected to increase by $\sim 80\%$ by 2100. Lower light availability might reduce phytoplankton photosynthetic rates and phytoplankton productivity. Szeligowska et al. (2021) found lower phytoplankton abundances in turbid and glacierinfluenced waters than in clear waters due to higher particle abundance. A model study by Polimene et al. (2022) on the Arctic shelf also found that primary productivity decreased by 4% to 16% with a 25% to 100% increase in terrestrial dissolved organic matter (DOM). The decrease in primary production was due to the interplay of light limitation and grazing pressure on primary producers. By attenuating underwater light, permafrost erosion may also shift the competitive balance between autotrophs and mixotrophs to favor the latter, with groups such as Dinophyceae and Cryptophyceae growing substantially better than pure autotrophs such as Bacillariophyceae (Szeligowska et al. 2020, 2021). Changes in phytoplankton community composition have already been observed in the Arctic Ocean around eastern Fram Strait, with chain-forming and large, centric diatoms such as Achnanthes taeniata, Navicula spp, Chaetoceros spp, and Thalassiosira spp being replaced by mixotrophic nanoflagellates and coccolithophorids due to warming and increased sediment inputs (Bauerfeind et al. 2009; Nöthig et al. 2015).

Apart from reducing the extent of the euphotic zone, low salinity from freshwater discharge might also negatively affect biodiversity. Brackish water tends to favor euryhaline plankton, leading to lower biodiversity when compared to marine water (Carmack et al. 2016; Hopwood et al. 2020). By increasing upper water column stratification, freshwater discharge also reduces vertical nutrient flux, consequently decreasing new production, and shifting the algal communities towards smaller cell sizes (Carmack et al. 2016). Smaller cells are more

effective in acquiring nutrients and less susceptible to gravitational settling than larger cells. In the Arctic Ocean, Li et al. (2009) found that phytoplankton with smaller cells thrived while larger cells languished due to increasing discharge of freshwater.

The elemental composition of autotrophs varies substantially as a function of carbon, solar energy, and mineral nutrient availabilities (Sterner and Elser 2003). Relative to nitrogen (N) and phosphorus (P), the higher input of terrigenous carbon might increase the elemental ratios of C : P and C : N in the Arctic coasts. This means that phytoplankton might be forced to utilize low cellular N and P quotas for growth sustenance, resulting in biomass with a disproportionate accumulation of C relative to N and P (Fig. 2). It remains unclear whether the negative effects of increased OC and light attenuation on primary production will be compensated by the positive influence of increased nutrient supply in the nearshore zones.

Bacterioplankton

Bacterioplankton have high carbon requirements and often rely on C-rich algal exudates for growth (Baines and Pace 1991). However, high inputs of allochthonous carbon can uncouple bacterial reliance on phytoplankton carbon exudates, a phenomenon that has been observed in high-latitude lakes (Tranvik 1988; Jansson et al. 2007). Hence, increased coastal erosion might stimulate bacterial production via greater labile C-rich terrestrial organic matter availability. Bacterioplankton and phytoplankton compete for nutrients such as nitrogen and phosphorus, and the expected negative influence of higher terrestrial C inputs on phytoplankton biomass may benefit bacterioplankton. By decreasing light availability and subsidizing with allochthonous C, elevated terrestrial inputs should favor heterotrophic bacterial production over autotrophic phytoplankton production (Sandberg et al. 2004; Figueroa et al. 2016). In support of this hypothesis, a 76% increase in freshwater discharge resulted in doubled the bacteria to phytoplankton production in the northern Baltic Sea due to inputs of colored humic substances and deeper mixed layers (Wikner and Andersson 2012). In another study, Meunier et al. (2017) observed that carbon enrichment decreases phytoplankton biomass and increases bacteria biomass, and suggested that bacteria outcompete phytoplankton where an alternate, readily bioavailable carbon source is accessible.

Studies indicate that about 30% of terrestrial DOC entering the Arctic coastal zones is readily bioavailable (Vonk et al. 2013, 2015; Mann et al. 2015). Beyond this carbon's lability and quality, the selective forces of these carbon compounds in structuring the bacterioplankton community should be considered (Gómez-Consarnau et al. 2012; Sosa et al. 2015). For instance, Niemi et al. (2014) found that DOM released from melting Arctic sea ice led to a 2.1–3.2-fold increase in bacterial cell size. Adding terrestrial DOM extracted



Fig. 2. Schematic illustration of the effects of terrestrial OC input from permafrost thaw and erosion on phytoplankton communities in the Arctic nearshore zone. The arrows indicate anticipated shifts in phytoplankton community structure and include changes in species composition, a decrease in cell size, changes in the elemental stoichiometry, and a reduction in phytoplankton abundance.

from the active permafrost layer to aquatic bacterial communities also shifts community composition by increasing the number of high nucleic acid-containing bacteria (Müller et al. 2018). This outcome may be due to the ability of larger bacteria to metabolize a broader spectrum of DOM than smaller bacteria (Grunert et al. 2021). While these shifts in bacterial communities may have large implications for overall carbon and nutrient cycling, these aspects remain largely unexplored in the coastal Arctic.

In short, increased coastal erosion and associated terrestrial DOC input benefit bacterioplankton and impair phytoplankton, which may affect higher trophic levels. The planktonic food web can be subdivided into two energy pathways, the photoautotrophic (phytoplankton-based) energy pathway, and the heterotrophic (bacterial-based) energy pathway (Azam et al. 1983; Meunier et al. 2016), which have different efficiencies (Fig. 3). In the following section, we explore the potential consequences of shifts in production at the base of the food chain for higher trophic levels.

Zooplankton

Phytoplankton-based food chains typically transfer carbon more efficiently to higher trophic levels than bacteria-based food chains (Berglund et al. 2007; Brett et al. 2009). However, it remains a fundamental question whether bacterial-derived carbon represents an addition to autotrophic energy sources or if it replaces it to some degree. As a potential supplement, a study by Degerman et al. (2018) in the Baltic Sea found an increase in zooplankton production due to the channeling of energy via both the phytoplankton and bacterial pathways. Similar changes may be expected in the Arctic when higher bacterial biomass dominates the heterotrophic food chain in the early spring months, before the phytoplankton bloom (Polimene et al. 2022). In another study by Karlsson et al. (2005) in highlatitude, unproductive lakes, they found a close association between the input of allochthonous OC sources and the input of limiting inorganic nitrogen and phosphorus. Bacteria have a higher affinity for inorganic nutrients than phytoplankton. In the northern Baltic Sea, Berglund et al. (2007) discovered that a noteworthy portion of the biomass in higher trophic levels may originate from terrestrial carbon sources, potentially mobilized by bacterioplankton. For instance, increased biomass of bacterioplankton could be advantageous for microzooplankton, such as certain ciliates known for their efficient consumption of bacteria (Berglund et al. 2007; Lefébure et al. 2013). Although microzooplankton is later consumed by mesozooplankton (Fig. 3), this additional trophic link reduces food web efficiency (Berglund et al. 2007; Jansson et al. 2007), as respiratory energy losses occur at each trophic transfer step, especially in cases where energy is limiting (Gan et al. 2023). Degerman et al. (2018) found minimal bacterial carbon transfer to consumers like mesozooplankton and fish.

Complicating their potential importance, bacterioplankton is generally considered poor-quality food for most mesozooplankton consumers because bacterioplankton lacks certain sterols and essential fatty acids (FAs; Brett and Müller-



Fig. 3. Conceptual model on Arctic planktonic food webs under present conditions and increased coastal erosion. In both figures, relative arrow width represents the amount of energy passed between links; relative organism size represents shifts in abundance. (**a**) When there is less terrestrial organic carbon (t-OC) input, the autotrophic energy pathway (black arrows) dominates. Energy mobilization for primary and secondary production comes from phytoplankton photosynthesis using solar radiation and dissolved inorganic carbon (DIC). Phytoplankton carbon and energy are then transferred to higher trophic levels. (**b**) Under increased coastal erosion, the heterotrophic energy pathway (orange arrow) dominates. Increased discharge of t-OC enhances bacterial production, which becomes a major source of energy mobilization in addition to phytoplankton-derived carbon. This energy is then transferred to the mesozooplankton (0.2–20 mm) and fish through phagotrophic zooplankton (< 200 μ m). However, much of the energy is dissipated due to additional links in the heterotrophic chain.

Navarra 1997; Martin-Creuzburg et al. 2011; Degerman et al. 2018). Mesozooplankton primarily obtain the essential highly unsaturated FAs, such as eicosapentaenoic acid and docosahexaenoic acid from phytoplankton they consume (Kattner and Hagen 2009). Feeding trials in a freshwater lake showed that *Daphnia* cannot survive on a diet containing only bacteria due to lack of these FAs (Martin-Creuzburg et al. 2011). Degerman et al. (2018) also found that fish production decreased due to the channeling of energy via the less efficient and longer bacterial pathway.

Suspended matter may also affect mesozooplankton filter and interception feeding (Margalef 1978), as well as grazing efficiency. Fine sediment loads in the water column can reduce the visibility of zooplankton. They are, therefore, forced to either invest additional energy in finding suitable food particles within the turbid water column or fill their gut with inorganic particles (Sommaruga 2015). Additional energy is also required to locate phytoplankton that sinks from the euphotic zone (Arendt et al. 2011), decreasing energy allocation to reproduction and somatic growth. Although suspended matter might shelter zooplankton from visual predators, it might also increase their specific weight through agglutination of cells, reducing their hunting activity (Margalef 1978; Szeligowska et al. 2021). To the best of our knowledge, no study to date has dealt with changes in the stoichiometric qualities and physiology of phytoplankton, bacteria, and zooplankton in the Arctic. Future studies should aim to fill this gap by manipulating primary producers' light and nutrient conditions to predict better how changes in food quality will influence zooplankton growth, development, and physiology in the Arctic.

POC from permafrost could also directly supplement food webs. For example, in freshwater lakes, POC has been shown

Iuma et al.

to contribute between 33% and 73% of the carbon incorporated into nearshore zooplankton, and between 20% and 50% of the carbon incorporated into fish (Cole et al. 2006). Even though some zooplankton species readily consume particulates, some studies found that high proportions of terrestrial POC in the diet of zooplankton from freshwater lakes reduce their growth and survival performance due to poor digestibility, and lack of essential biochemicals (FAs) and minerals (P) (Wenzel et al. 2012, 2021). Our current understanding of how zooplankton in marine coastal habitats will respond to OC from permafrost is severely hindered by the lack of studies on these ecosystems, and we recognize that some biogeochemical processes in marine systems might differ from freshwater habitats. However, similar biological processes such as increased heterotrophy under higher OC, have been observed in both systems. We can therefore hypothesize that elevated permafrost OC will favor heterotrophic energy pathways, which will have negative consequences for mesozooplankton biomass, and may further impact fish if it replaces the autotrophic energy pathway. Studies that disentangle and combine assessments of Arctic primary productivity under varying carbon concentrations and light intensities can better elucidate the primary drivers of biological shifts in the microbial food web and the potential impacts on higher trophic communities.



- · Increase in active layer depth
- · Thawing of coastal sediment
- · Increase of coastal erosion
- · Reduction in light conditions
- · Increase in C input, relative to N and P
- · Loss of fishing grounds
- · Increase in bacterial production
- · Shifts to heterotrophic bacterial-based energy pathway
- Additional trophic steps in the food web

Fig. 4. Potential impacts of increased permafrost thaw and erosion on coastal biogeochemical processes. (1) Increased warming might lead to an increase in active layer depth and thawing of permafrost sediment at the coast. (2) In the nearshore zone, increased discharge of sediment nutrients and carbon might enhance turbidity in the nearshore zones, as well as higher C input, relative to N and P. Additionally, coastal erosion might lead to loss of fishing ground, as well as migration to offshore areas with lower turbidity. (3) Higher turbidity and C input might support bacterial production shifting the energy pathway to a heterotrophic bacterial-based pathway. Such shifts might lead to longer trophic links for zooplankton and fish. Figure adapted from Fritz et al. (2017), and modified by Yves Nowak (AWI).

Juma et al.

Conclusions and future perspectives

Climate warming is particularly rapid in the Arctic, accelerating the thawing and erosion of permafrost. Elevated erosion of permafrost material along the Arctic coast is expected to increase the overall coastal area influenced by terrestrial inputs. The seawater in these areas will become more turbid and have higher dissolved and particulate carbon content. We expect that this darkening will reduce phytoplankton productivity due to light limitation, and alter plankton species composition, and biomass (Fig. 4).

The input of t-DOM might also supply bacteria with a direct energy source, enhancing bacterial production and the heterotrophic energy pathway. In this case, the phytoplankton energy pathway might be replaced to some degree in the coastal marine habitats. Although bacterial energy might be a supplement for some organisms, we suggest that the effects of low nutrient-use efficiency and low energy transfer efficiency induced by heterotrophic energy mobilization will negatively affect size-structured dynamics among zooplankton and subsequently reduce food availability for fish.

To date, literature on coastal erosion dynamics and the associated effects on nearshore food webs is characterized by small datasets. It remains an open question of how different permafrost coasts (stable and degraded coasts) influence carbon and nutrient input, as well as the phytoplankton and zooplankton community structures in the nearshore zones. Additionally, it has been demonstrated that organisms at higher trophic levels can be supported by either phytoplankton or bacterial energy pathways, depending on the intensity of carbon input. Isotopic analysis can elucidate the sources of OC and its transfer through the trophic chain. Moreover, no studies in the Arctic have directly compared zooplankton growth, reproduction, and survival, when fed with bacterial, algal, or t-DOM-dominated diets. Experimental studies with manipulated quantities of light and carbon at different trophic levels will help predict the potential effects of increasing erosion on both top-down and bottom-up processes. Clarifications on these aspects should substantially enhance our understanding of the manifold ways the Arctic ecosystem is changing in response to Arctic warming, and thus guide Arctic coastal management towards a greater focus on sustainable nearshore ecosystem conservation.

References

- Ardyna, M., M. Babin, E. Devred, A. Forest, M. Gosselin, P. Raimbault, and J. Tremblay. 2017. Shelf-basin gradients shape ecological phytoplankton niches and community composition in the coastal Arctic Ocean (Beaufort Sea). Limnol. Oceanogr. 62: 2113–2132. doi:10.1002/lno.10554
- Arendt, K. E., J. Dutz, S. H. Jónasdóttir, S. Jung-Madsen, J. Mortensen, E. F. Møller, and T. G. Nielsen. 2011. Effects of suspended sediments on copepods feeding in a glacial

influenced sub-Arctic fjord. J. Plankton Res. **33**: 1526–1537. doi:10.1093/plankt/fbr054

- Atkinson, D. 2005. Observed storminess patterns and trends in the circum-Arctic coastal regime. Geo-Mar. Lett. **25**: 98– 109. doi:10.1007/s00367-004-0191-0
- Azam, F., T. Fenchel, J. Field, J. Gray, L. Meyer-Reil, and F. Thingstad. 1983. The ecological role of water-column microbes in the sea. Mar. Ecol. Prog. Ser. 10: 257–263. doi:10.3354/meps010257
- Baines, S. B., and M. L. Pace. 1991. The production of dissolved organic matter by phytoplankton and its importance to bacteria: Patterns across marine and freshwater systems. Limnol. Oceanogr. 36: 1078–1090. doi:10.4319/ lo.1991.36.6.1078
- Barker, J. D., M. J. Sharp, S. J. Fitzsimons, and R. J. Turner. 2006. Abundance and dynamics of dissolved organic carbon in glacier systems. Arct. Antarct. Alp. Res. 38: 163–172. doi:10.1657/1523-0430(2006)38[163:AADODO]2.0.CO;2
- Bauerfeind, E., and others. 2009. Particle sedimentation patterns in the eastern Fram Strait during 2000–2005: Results from the Arctic long-term observatory HAUSGARTEN. Deep-Sea Res. II Oceanogr. Res. Pap. 56: 1471–1487. doi: 10.1016/j.dsr.2009.04.011
- Berglund, J., U. Müren, U. Båmstedt, and A. Andersson. 2007.
 Efficiency of a phytoplankton-based and a bacterial-based food web in a pelagic marine system. Limnol. Oceanogr. 52: 121–131. Portico. doi:10.4319/lo.2007.52.1.0121
- Bluhm, A. B., and G. Rolf. 2008. Regional variability in food availability. Ecol. Appl. 18: S77–S96. doi:10.3389/feart. 2021.640580
- Brett, M. T., and D. C. Müller-Navarra. 1997. The role of highly unsaturated fatty acids in aquatic foodweb processes. Freshw. Biol. **38**: 483–499. doi:10.1046/j.1365-2427. 1997.00220.x
- Brett, M. T., M. J. Kainz, S. J. Taipale, and H. Seshan. 2009. Phytoplankton, not allochthonous carbon, sustains herbivorous zooplankton production. Proc. Natl. Acad. Sci. U. S. A. **106**: 21197–21201. doi:10.1073/pnas.0904129106
- Brett, M. T., and others. 2017. How important are terrestrial organic carbon inputs for secondary production in freshwater ecosystems? Freshw. Biol. **62**: 833–853. doi:10.1111/fwb.12909
- Calbet, A. 2008. The trophic roles of microzooplankton in marine systems. ICES J. Mar. Sci. **65**: 325–331. doi:10.109 3/icesjms/fsn013
- Carmack, E. C., and others. 2016. Freshwater and its role in the Arctic marine system: Sources, disposition, storage, export, and physical and biogeochemical consequences in the Arctic and global oceans. J. Geophys. Res. Biogeo. **121**: 675–717. doi:10.1002/2015JG003140
- Casas-Prat, M., and X. L. Wang. 2020. Sea ice retreat contributes to projected increases in extreme Arctic Ocean surface waves. Geophys. Res. Lett. **47**. doi:10.1029/2020GL 088100

Juma et al.

- Cohen, J., and others. 2014. Recent Arctic amplification and extreme mid-latitude weather. Nat. Geosci. **7**: 627–637. doi:10.1038/ngeo2234
- Cole, J. J., S. R. Carpenter, M. L. Pace, M. C. Van De Bogert, J. L. Kitchell, and J. R. Hodgson. 2006. Differential support of lake food webs by three types of terrestrial organic carbon. Ecol. Lett. **9**: 558–568. doi:10.1111/j.1461-0248.2006. 00898.x
- Degerman, R., R. Lefébure, P. Byström, U. Båmstedt, S. Larsson, and A. Andersson. 2018. Food web interactions determine energy transfer efficiency and top consumer responses to inputs of dissolved organic carbon. Hydrobiologia **805**: 131–146. doi:10.1007/s10750-017 -3298-9
- Farquharson, L. M., D. H. Mann, D. K. Swanson, B. M. Jones, R. M. Buzard, and J. W. Jordan. 2018. Temporal and spatial variability in coastline response to declining sea-ice in northwest Alaska. Mar. Geol. **404**: 71–83. doi:10.1016/j. margeo.2018.07.007
- Figueroa, D., O. F. Rowe, J. Paczkowska, C. Legrand, and A. Andersson. 2016. Allochthonous carbon—A major driver of bacterioplankton production in the subarctic northern Baltic Sea. Microb. Ecol. **71**: 789–801. doi:10.1007/s00248 -015-0714-4
- Fritz, M., T. Opel, G. Tanski, U. Herzschuh, H. Meyer, A. Eulenburg, and H. Lantuit. 2015. Dissolved organic carbon (DOC) in Arctic ground ice. Cryosphere 9: 737–752. doi:10. 5194/tc-9-737-2015
- Fritz, M., J. E. Vonk, and H. Lantuit. 2017. Collapsing Arctic coastlines. Nat. Clim. Change 7: 6–7. doi:10.1038/nclima te3188
- Galappaththi, E. K., J. D. Ford, E. M. Bennett, and F. Berkes. 2019. Climate change and community fisheries in the arctic: A case study from Pangnirtung, Canada. J. Environ. Manage. **250**: 109534. doi:10.1016/j.jenvman.2019. 109534
- Gan, Y., Y. Su, and J. Ma. 2023. Effects of microbial-converted ancient permafrost organic carbon on the growth and reproduction of *Daphnia magna*. Oecologia **203**: 335–348. doi:10.1007/s00442-023-05467-x
- Golubeva, E., G. Platov, D. Iakshina, and M. Kraineva. 2019. A simulated distribution of Siberian river runoff in the Arctic Ocean. IOP Conf. Ser. Earth Environ. Sci. **386**: 012022. doi: 10.1088/1755-1315/386/1/012022
- Gómez-Consarnau, L., M. V. Lindh, J. M. Gasol, and J. Pinhassi. 2012. Structuring of bacterioplankton communities by. Environ. Microbiol. **14**: 2361–2378. doi:10.1111/j. 1462-2920.2012.02804.x
- Grotheer, H., and others. 2020. Burial and origin of permafrost-derived carbon in the nearshore zone of the Southern Canadian Beaufort Sea. Geophys. Res. Lett. **47**. doi:10. 1029/2019GL085897
- Grunert, B. K., M. Tzortziou, P. Neale, A. Menendez, and P. Hernes. 2021. DOM degradation by light and microbes

along the Yukon River-coastal ocean continuum. Sci. Rep. **11**: 1–14. doi:10.1038/s41598-021-89327-9

- Hawkings, J., and others. 2016. The Greenland Ice Sheet as a hot spot of phosphorus weathering and export in the Arctic. Global Biogeochem. Cycles **30**: 191–210. doi:10.1002/ 2015GB005237
- Hopwood, M. J., and others. 2020. Review article: How does glacier discharge affect marine biogeochemistry and primary production in the Arctic? Cryosphere **14**: 1347–1383. doi:10.5194/tc-14-1347-2020
- Hugelius, G., J. Strauss, S. Zubrzycki, J. W. Harden, E. A. G. Schuur, C. Ping, and L. Schirrmeister. 2014. Estimated stocks of circumpolar permafrost carbon with quantified uncertainty ranges and identified data gaps. Biogeosciences 11: 6573–6593. doi:10.5194/bg-11-6573-2014
- Irrgang, A. M., and others. 2022. Drivers, dynamics and impacts of changing Arctic coasts. Nat. Rev. Earth Environ. 3: 39–54. doi:10.1038/s43017-021-00232-1
- Jansson, M., L. Persson, A. M. De Roos, R. I. Jones, and L. J. Tranvik. 2007. Terrestrial carbon and intraspecific size-variation shape lake ecosystems. Trends Ecol. Evol. 22: 316– 322. doi:10.1016/j.tree.2007.02.015
- Jong, D., L. Bröder, G. Tanski, M. Fritz, and H. Lantuit. 2020. Nearshore zone dynamics determine pathway of organic carbon from eroding permafrost coasts. Geophys. Res. Lett. 47: e2020GL088561. doi:10.1029/2020GL088561
- Karlsson, J., A. Jonsson, and M. Jansson. 2005. Productivity of high-latitude lakes: Climate effect inferred from altitude gradient. Glob. Chang. Biol. **11**: 710–715. doi:10.1111/j. 1365-2486.2005.00945.x
- Kattner, G., and W. Hagen. 2009. Lipids in marine copepods: Latitudinal characteristics and perspective to global warming, p. 257–280. *In, Lipids in aquatic ecosystems*. Springer. doi:10.1007/978-0-387-89366-2_11
- Kupilik, M., M. Ulmgren, and D. Brunswick. 2020. Bayesian parameter estimation for Arctic coastal erosion under the effects of climate change. IEEE J. Sel. Top. Appl. Earth Obs. Remote Sens. 13: 3595–3604. doi:10.1109/JSTARS.2020.3004291
- Lantuit, H., and others. 2012. The Arctic coastal dynamics database: A new classification scheme and statistics on Arctic permafrost coastlines. Estuaries Coast **35**: 383–400. doi: 10.1007/s12237-010-9362-6
- Lefébure, R., R. Degerman, A. Andersson, S. Larsson, L. O. Eriksson, U. Båmstedt, and P. Byström. 2013. Impacts of elevated terrestrial nutrient loads and temperature on pelagic food-web efficiency and fish production. Glob. Chang. Biol. **19**: 1358–1372. doi:10.1111/gcb.12134
- Li, W. K. W., F. A. McLaughlin, C. Lovejoy, and E. C. Carmack. 2009. Smallest algae thrive as the Arctic Ocean freshens. Science **326**: 539. doi:10.1126/science.1179798
- Lim, M., D. Whalen, J. Martin, P. J. Mann, S. Hayes, P. Fraser, H. B. Berry, and D. Ouellette. 2020. Massive ice control on permafrost coast erosion and sensitivity. Geophys. Res. Lett. 47. doi:10.1029/2020GL087917

- Lindemann, C., Ø. Fiksen, K. H. Andersen, and D. L. Aksnes. 2016. Scaling laws in phytoplankton nutrient uptake affinity. Front. Mar. Sci. **3**: 1–6. doi:10.3389/fmars.2016.00026
- Mann, P. J., T. I. Eglinton, C. P. McIntyre, N. Zimov, A. Davydova, J. E. Vonk, R. M. Holmes, and R. G. M. Spencer. 2015. Utilization of ancient permafrost carbon in headwaters of Arctic fluvial networks. Nat. Commun. 6: 7856. doi: 10.1038/ncomms8856
- Mann, P. J., and others. 2021. Degrading permafrost river catchments and their impact on Arctic Ocean nearshore processes. Ambio **51**: 439–455. doi:10.1007/s1328 0-021-01666-z
- Margalef, R. 1978. Life-forms of phytoplankton as survival alternatives in an unstable environment. Oceanol. Acta 1: 493–509. https://archimer.ifremer.fr/doc/00123/23403/21230.pdf
- Martin-Creuzburg, D., B. Beck, and H. M. Freese. 2011. Food quality of heterotrophic bacteria for *Daphnia magna*: Evidence for a limitation by sterols. FEMS Microbiol. Ecol. **76**: 592–601. doi:10.1111/j.1574-6941.2011.01076.x
- McClelland, J. W., R. M. Holmes, K. H. Dunton, and R. W. Macdonald. 2012. The Arctic Ocean estuary. Estuaries Coast **35**: 353–368. doi:10.1007/s12237-010-9357-3
- McGovern, M., A. K. Pavlov, A. Deininger, M. A. Granskog, E. Leu, J. E. Søreide, and A. E. Poste. 2020. Terrestrial inputs drive seasonality in organic matter and nutrient biogeochemistry in a high Arctic fjord system (Isfjorden, Svalbard). Front. Mar. Sci. **7**: 1–15. doi:10.3389/fmars.2020. 542563
- Mcguire, A. D., and others. 2009. Sensitivity of the carbon cycle in the Arctic to climate change. Ecological monographs **79**: 523–555. doi:10.1890/08-2025.1
- Meunier, C. L., M. J. Gundale, I. S. Sánchez, and A. Liess. 2016. Impact of nitrogen deposition on forest and lake food webs in nitrogen-limited environments. Glob. Chang. Biol. 22: 164–179. doi:10.1111/gcb.12967
- Meunier, C. L., A. Liess, A. Andersson, S. Brugel, J. Paczkowska, H. Rahman, B. Skoglund, and O. F. Rowe. 2017. Allochthonous carbon is a major driver of the microbial food web—A mesocosm study simulating elevated terrestrial matter runoff. Mar. Environ. Res. **129**: 236–244. doi:10.1016/j.marenvres.2017.06.008
- Müller, O., L. Seuthe, G. Bratbak, and M. L. Paulsen. 2018. Bacterial response to permafrost derived organic matter input in an Arctic Fjord. Front. Mar. Sci. 5. doi:10.3389/ fmars.2018.00263
- Mundy, C. J., and others. 2009. Contribution of under-ice primary production to an ice-edge upwelling phytoplankton bloom in the Canadian Beaufort Sea. Geophys. Res. Lett. **36**: 1–5. doi:10.1029/2009GL038837
- Murphy, G. E. P., T. N. Romanuk, and B. Worm. 2020. Cascading effects of climate change on plankton community structure. Ecol. Evol. **2019**: 2170–2181. doi:10.1002/ ece3.6055

- Negrete-García, G., J. Y. Luo, M. C. Long, K. Lindsay, M. Levy, and A. D. Barton. 2022. Plankton energy flows using a global size-structured and trait-based model. Prog. Oceanogr. **209**: 102898. doi:10.1016/j.pocean.2022.102898
- Nielsen, D. M., P. Pieper, A. Barkhordarian, P. Overduin, T. Ilyina, V. Brovkin, J. Baehr, and M. Dobrynin. 2022. Increase in Arctic coastal erosion and its sensitivity to warming in the twenty-first century. Nat. Clim. Change 12: 263–270. doi:10.1038/s41558-022-01281-0
- Niemi, A., G. Meisterhans, and C. Michel. 2014. Response of under-ice prokaryotes to experimental sea-ice DOM enrichment. Aquat. Microb. Ecol. **73**: 17–28. doi:10.3354/ame 01706
- Nöthig, E., and others. 2015. Summertime plankton ecology in Fram Strait—A compilation of long- and short-term observations. Polar Res. **34**. doi:10.3402/polar.v34.23349
- Overduin, P. P., M. C. Strzelecki, M. N. Grigoriev, N. Couture, H. Lantuit, D. St-Hilaire-Gravel, F. Gunther, and S. Wetterich. 2014. Coastal changes in the Arctic, p. 103–129. *In P. Martini and H. Wanless [eds.], Sedimentary coastal zones from high to low latitudes: Similarities and differences.* Geological Society of London. doi:10.1144/SP388.13
- Park, J. H., S.-J. Kim, H.-G. Lim, J.-S. Kug, E. J. Yang, and B.-M. Kim. 2023. Phytoplankton responses to increasing Arctic river discharge under the present and future climate simulations OPEN ACCESS Phytoplankton responses to increasing Arctic river discharge under the present and future climate simulations. Environ. Res. Lett. 18. doi:10.1088/17 48-9326/acd568
- Peterson, B. J., R. M. Holmes, J. W. McClelland, C. J. Vörösmarty, R. B. Lammers, A. I. Shiklomanov, I. A. Shiklomanov, and S. Rahmstorf. 2002. Increasing river discharge to the Arctic Ocean. Science **298**: 2171–2173. doi: 10.1126/science.1077445
- Piontek, J., L. Galgani, E. M. Nöthig, I. Peeken, and A. Engel. 2021. Organic matter composition and heterotrophic bacterial activity at declining summer sea ice in the central Arctic Ocean. Limnol. Oceanogr. 66: S343–S362. doi:10. 1002/lno.11639
- Polimene, L., R. Torres, H. R. Powley, M. Bedington, B. Juhls, J. Palmtag, J. Strauss, and P. J. Mann. 2022. Biological lability of terrestrial DOM increases CO₂ outgassing across Arctic shelves. Biogeochemistry **160**: 289–300. doi:10.1007/ s10533-022-00961-5
- Popova, E. E., A. Yool, A. C. Coward, Y. K. Aksenov, S. G. Alderson, B. A. De Cuevas, and T. R. Anderson. 2010. Control of primary production in the Arctic by nutrients and light: Insights from a high resolution ocean general circulation model. Biogeosciences **7**: 3569–3591. doi:10.5194/ bg-7-3569-2010
- Ramage, J., L. Jungsberg, S. Wang, S. Westermann, H. Lantuit, and T. Heleniak. 2021. Population living on permafrost in the Arctic. Popul. Environ. **43**: 22–38. doi:10.1007/ s11111-020-00370-6

- Rantanen, M., A. Y. Karpechko, A. Lipponen, K. Nordling, O. Hyvärinen, K. Ruosteenoja, T. Vihma, and A. Laaksonen. 2022. The Arctic has warmed nearly four times faster than the globe since 1979. Commun. Earth Environ. 3: 1–10. doi:10.1038/s43247-022-00498-3
- Sandberg, J., A. Andersson, S. Johansson, and J. Wikner. 2004. Pelagic food web structure and carbon budget in the northern Baltic Sea : Potential importance of terrigenous carbon. Mar. Ecol. Prog. Ser. 268: 13–29. doi:10.3354/meps268013
- Schmoker, C., S. Hernández-León, and A. Calbet. 2013. Microzooplankton grazing in the oceans: Impacts, data variability, knowledge gaps and future directions. J. Plankton Res. 35: 691–706. doi:10.1093/plankt/fbt023
- Singh, R. K., A. Vader, C. J. Mundy, J. E. Søreide, K. Iken, K. H. Dunton, L. Castro de la Guardia, M. K. Sejr, and S. Bélanger. 2022. Satellite-derived photosynthetically available radiation at the coastal Arctic seafloor. Remote Sens. (Basel) 14: 1–33. doi:10.3390/rs14205180
- Sipler, R. E., C. T. E. Kellogg, T. L. Connelly, Q. N. Roberts, P. L. Yager, and D. A. Bronk. 2017. Microbial community response to terrestrially derived dissolved organic matter in the coastal Arctic. Front. Microbiol. 8: 1–19. doi:10.3389/ fmicb.2017.01018
- Slater, A. G., and D. M. Lawrence. 2013. Diagnosing present and future permafrost from climate models. J. Climate 26: 5608–5623. doi:10.1175/JCLI-D-12-00341.1
- Sommaruga, R. 2015. When glaciers and ice sheets melt: Consequences for planktonic organisms. J. Plankton Res. **37**: 509–518. doi:10.1093/plankt/fbv027
- Sosa, O. A., S. M. Gifford, D. J. Repeta, and E. F. Delong. 2015. High molecular weight dissolved organic matter enrichment selects for methylotrophs in dilution to extinction cultures. Int. Soc. Microb. Ecol. 9: 2725–2739. doi:10.1038/ ismej.2015.68
- Sterner, R. W., and J. J. Elser. 2003. Ecological stoichiometry: The biology of elements from molecules to the biosphere. Princeton Univ. Press. doi:10.1515/9781400885695
- Szeligowska, M., E. Trudnowska, R. Boehnke, A. M. Dąbrowska, J. M. Wiktor, S. Sagan, and K. Błachowiak-Samołyk. 2020. Spatial patterns of particles and plankton in the warming Arctic Fjord (Isfjorden, west Spitsbergen) in seven consecutive mid-summers (2013–2019). Front. Mar. Sci. 7: 1–18. doi:10.3389/fmars.2020.00584
- Szeligowska, M., E. Trudnowska, R. Boehnke, A. M. Dąbrowska, K. Dragańska-Deja, K. Deja, M. Darecki, and K. Błachowiak-Samołyk. 2021. The interplay between plankton and particles in the Isfjorden waters influenced by marine- and land-terminating glaciers. Sci. Total Environ. 780: 146491. doi:10.1016/j.scitotenv.2021.146491
- Tank, S. E., M. Manizza, R. M. Holmes, J. W. McClelland, and B. J. Peterson. 2012. The processing and impact of dissolved riverine nitrogen in the Arctic Ocean. Estuaries Coast 35: 401–415. doi:10.1007/s12237-011-9417-3

- Tanski, G., C. Couture, H. Lantuit, A. Eulenburg, and M. Fritz. 2016. Eroding permafrost coasts release low amounts of dissolved organic carbon (DOC) from ground ice into the nearshore zone of the Arctic Ocean. Global Biogeochem. Cycles **30**: 1054–1068. doi:10.1002/2015GB005337
- Terhaar, J., J. C. Orr, C. Ethe, P. Regnier, and L. Bopp. 2019. Simulated Arctic Ocean response to doubling of riverine carbon and nutrient delivery. Global Biogeochem. Cycles 33: 1048–1070. doi:10.1029/2019GB006200
- Terhaar, J., R. Lauerwald, P. Regnier, N. Gruber, and L. Bopp. 2021. Around one third of current Arctic Ocean primary production sustained by rivers and coastal erosion. Nat. Commun. **12**: 1–10. doi:10.1038/s41467-020-20470-z
- Tanski, G., and others. 2021. Permafrost carbon and CO₂ pathways differ at contrasting coastal erosion sites in the Canadian Arctic. Front. Earth Sci. **9**: 1–20. doi:10.3389/feart.2021.630493
- Thompson, M. S., F. J. Wrona, and T. D. Prowse-. 2012. Shifts in plankton, nutrient and light relationships in small tundra lakes caused by localized permafrost thaw. Arctic **65**: 367–376. doi:10.14430/arctic4235
- Tranvik, L. J. 1988. Availability of dissolved organic carbon for planktonic bacteria in oligotrophic lakes of differing humic content. Microb. Ecol. 16: 311–322. doi:10.1007/BF02011702
- Tremblay, J. É., L. G. Anderson, P. Matrai, P. Coupel, S. Bélanger, C. Michel, and M. Reigstad. 2015. Global and regional drivers of nutrient supply, primary production and CO₂ drawdown in the changing Arctic Ocean. Prog. Oceanogr. **139**: 171–196. doi:10.1016/j.pocean.2015.08.009
- Underwood, G. J. C., C. Michel, G. Meisterhans, A. Niemi, C. Belzile, M. Witt, A. J. Dumbrell, and B. P. Koch. 2019. Organic matter from Arctic sea-ice loss alters bacterial community structure and function. Nat. Clim. Change 9: 170–176. doi:10.1038/s41558-018-0391-7
- Vonk, J. E., and others. 2013. High biolability of ancient permafrost carbon upon thaw. Geophys. Res. Lett. 40: 2689– 2693. doi:10.1002/grl.50348
- Vonk, J. E., S. E. Tank, P. J. Mann, R. G. M. Spencer, C. C. Treat, R. G. Striegl, B. W. Abbott, and K. P. Wickland. 2015. Biodegradability of dissolved organic carbon in permafrost soils and aquatic systems: A meta-analysis. Biogeosciences 12: 6915–6930. doi:10.5194/bg-12-6915-2015
- Vrede, T., and others. 2011. Ecological stoichiometry of Eurasian perch—Intraspecific variation due to size, habitat and diet. Oikos **120**: 886–896. doi:10.1111/j.1600-0706.2010.18939.x
- Wegner, C., and others. 2015. Variability in transport of terrigenous material on the shelves and the deep Arctic Ocean during the Holocene. Polar Res. **34**. doi:10.3402/polar.v34. 24964
- Wenzel, A., A. K. Bergström, M. Jansson, and T. Vrede. 2012. Poor direct exploitation of terrestrial particulate organic material from peat layers by *Daphnia galeata*. Can. J. Fish. Aquat. Sci. 69: 1870–1880. doi:10.1139/f2012-110

- Wenzel, A., T. Vrede, M. Jansson, and A. K. Bergström. 2021. Daphnia performance on diets containing different combinations of high-quality algae, heterotrophic bacteria, and allochthonous particulate organic matter. Freshw. Biol. 66: 157–168. doi:10.1111/fwb.13626
- Wikner, J., and A. Andersson. 2012. Increased freshwater discharge shifts the trophic balance in the coastal zone of the northern Baltic Sea. Glob. Chang. Biol. **18**: 2509–2519. doi: 10.1111/j.1365-2486.2012.02718.x
- Zeller, D., S. Booth, E. Pakhomov, W. Swartz, and D. Pauly. 2011. Arctic fisheries catches in Russia, USA, and Canada: Baselines for neglected ecosystems. Polar Biol. **34**: 955–973. doi:10.1007/s00300-010-0952-3

Acknowledgments

Gabriel A. Juma was supported by the Deutscher Akademischer Austauschdienst (DAAD). Caroline Weber was supported by the Alfred Wegener Institute-International Science Program for Interactive Research in Earth Systems (AWI INSPIRES). The authors acknowledge Open-access funding provided by the Alfred Wegener Institute.

Conflict of Interest

None declared.

Submitted 02 February 2024 Revised 11 October 2024 Accepted 15 October 2024